Author's response to reviews

Title: Linezolid-induced black hairy tongue: a case report

Authors:

Faisal A Khasawneh (faisal.khasawneh@ttuhsc.edu)
Dereje F Moti (Dereje.moti@ttuhsc.edu)
Joseph A Zorek (Joseph.zorek@ttuhsc.edu)

Version: 2 Date: 15 November 2012

Author's response to reviews: see over
Dear editor:

The suggested changes have been undertaken as below:

1) Please include the study design in your title, i.e. Case report. For example: A presenting with B in C: a case report.

The title has been changed to include the study design. Now, it reads” Linezolid-induced black hairy tongue: a case report “

2) Please include the ethnicity of the patient in the abstract and in the main case presentation section of the manuscript. (Eg: Caucasian)

The patient ethnicity has been added to abstract and case presentation. Please, see abstract (page 2, line 3) and main case presentation (page 3, line 14).

Thank you for the opportunity.

Yours sincerely,

Faisal A. Khasawneh, MD
Assistant Professor, Infectious diseases
Department of Internal Medicine
Texas Tech University Health Sciences Center
1400 S. Coulter Street
Amarillo, TX 79106
E-mail: faisal.khasawneh@ttuhsc.edu
Phone: 806-354-5480
Fax: 806-354-5765